Author's response to reviews

Title: Improving organisational systems for diabetes care in Australian Indigenous communities

Authors:

Ross Bailie (ross.bailie@menzies.edu.au)
Damin Si (damin.si@menzies.edu.au)
Michelle Dowden (michelle.dowden@menzies.edu.au)
Lynette O'Donoghue (lynette.o.donoghue@menzies.edu.au)
Christine Connors (christine.connors@nt.gov.au)
Gary Robinson (gary.robinson@cdu.edu.au)
Joan Cunningham (joan.cunningham@menzies.edu.au)
Tarun Weeramanthri (tarun.weeramanthri@nt.gov.au)

Version: 3 Date: 4 May 2007

Author's response to reviews:

Dear Editor,

We have made formatting changes as follows:

1. All references in the text have been placed before the full stop.

2. Figure titles and legends have been placed after the reference list, and figures themselves have been deleted from the manuscript. Figures have been uploaded as separate files.

3. We are unable to improve the image resolution of figure 1, but we believe it is clear enough to show the key information.

With best wishes,

Ross Bailie